Author's response to reviews

Title: The relationship between school physical activity policy and objectively measured physical activity of elementary school students: A multilevel model analysis

Authors:

Guy Faulkner (guy.faulkner@utoronto.ca)
Laura Zeglen (laura.zeglen@utoronto.ca)
Scott Leatherdale (sleather@uwaterloo.ca)
Steve Manske (steve.manske@uwaterloo.ca)
Michelle Stone (michelle.stone@dal.ca)

Version: 4 Date: 25 March 2014

Author's response to reviews:

it should be #2μ0 in the online abstract but it keeps changing this. I have changed it to 'Intraclass correlation' if easier. All correct throughout text.

thanks

guy